Author's response to reviews

Title: Short course daily prednisolone therapy at the time of upper respiratory tract infection in children with relapsing steroid sensitive nephrotic syndrome (PREDNOS 2): protocol for a randomised controlled trial

Authors:

Nicholas J A Webb (nicholas.webb@cmft.nhs.uk)
Emma Frew (e.frew@bham.ac.uk)
Elizabeth A Brettell (e.a.brettell@bham.ac.uk)
David V Milford (david.milford@bch.nhs.uk)
Detlef Bockenhauer (detlef.bockenhauer@nhs.net)
Moin A Saleem (moinsaleem@hotmail.com)
Martin Christian (martin.christian@nuh.nhs.uk)
Angela S Hall (angela.hall@uhl-tr.nhs.uk)
Ania Koziell (ania.koziell@kcl.ac.uk)
Heather Maxwell (heather.maxwell@yorkhill.scot.nhs.uk)
Shivram Hegde (shavaram.hegde@wales.nhs.uk)
Eric Finlay (eric.finlay@nhs.net)
Rodney D Gilbert (rodney.gilbert@uhs.nhs.uk)
Jenny Booth (jenny.booth@nuth.nhs.uk)
Caroline Jones (caroline.jones@alderhey.nhs.uk)
Karl McKeever (karl.mckeever@belfasttrust.hscni.nhs.uk)
Wendy Cook (wedycook63@tesco.net)
Natalie J Ives (n.j.ives@bham.ac.uk)

Version: 2
Date: 18 October 2013

Author's response to reviews:

Dear Trials

Thank you for your email dated 10 10 13. I have now added information regarding ethical committee approval and current trial status. I have also added the figure legend after the reference list.

I have emailed you documents confirming funding and ethical approval.

I hope that the manuscript now meets with your approval.

Yours sincerely

Professor Nick Webb